CS 190 Final Course Evaluation Handout #8 April 30, 1999

We would like to get feedback on how the course went so that we can make improvements for future years.

What’s the most useful thing you learned?

Is the course content appropriate? Would you suggest changes?

Is the course organization appropriate? Would you suggest changes?

How beneficial are lectures/discussions to the project?

To your development as a software engineer?

Would you change the allocation of classes to lectures/discussions, class presentations, and group meetings?

Do you have enough access to the instructor and TAs?

Any feedback on the texts?

Can you suggest ways for the teacher to be more effective?

The TAs?

Any other feedback?